Author’s response to reviews

Title: Modified double patch repair with infarct exclusion technique for ventricular septal perforation: A case study

Authors:

Takuma Yamasaki (takuma360j@gmail.com)
Shuhei Fujita (fujishu0615@gmail.com)
Yuji Kaku (yuji_1350@ybb.ne.jp)
Junko Katagiri (junkokata517@gmail.com)
Takeshi Hiramatsu (takhiramatsu@yahoo.co.jp)

Version: 1 Date: 13 Jan 2018

Author’s response to reviews:

January 13, 2018

Dear Afzal Zaidi, MBBChir, FRCS(CTh) and El Kapetanakis Reviewers

Dear Dr Vipin Zamvar

Editors
Journal of Cardiothoracic Surgery

Thank you not only for carefully reading our case report but also for properly evaluating it. We wish to revise a case report for publication in Journal of Cardiothoracic Surgery titled

“Modified double patch repair with infarct exclusion technique for ventricular septal perforation: A case study”.

We were asked to improve the English in the manuscript, so we revised the manuscript using a reputable English editing service. We used Editage(www.editage.jp) for English language editing. There is no change in the content of this paper.

We believe that this paper will be of interest to the readership of your journal because of the potential clinical and research applications of our findings and report.
Thank you for your consideration. I look forward to hearing from you.

Sincerely,

Takuma Yamasaki